
Daily Status Mon. 15-Oct-12
15-Oct-12

Personnel: JohnM.,  TomH arriving today.

Weather&Ops:
Remained mostly clear last night with light wind.    Clear sky in am and seasonal temps.   Mostly calm throughout the day and evening.

Sensors:
Overall things looking good.   Nearly everything is reporting except the Picarro, which is still awaiting to be repaired at their factory.  

CSATs still having diag hits this morning. 

Krypton voltages have sagged and need to be cleaned.

TRH's / fans all seem ok and everyone reporting.   There was a cable problem with at least one of the trh's at a17, see trh entry.

TRH connector maintenance: cleaned and coated trh binders at sites 9-20 causing outages.   See other log entry for times.
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